Author's response to reviews

Title: Use of the bootstrap in analysing cost data from cluster randomised trials: some simulation results

Authors:

Terry N Flynn (terry.flynn@bristol.ac.uk)
Tim J Peters (tim.peters@bristol.ac.uk)

Version: 3 Date: 25 October 2004

Author's response to reviews:

We have revised the paper in the light of the formatting changes requested. The changes made are detailed below.

Word count - word count removed from the title page.

References - In the manuscript text, references placed before the full stop.

Section headings - italics deleted.

Competing interests - New BMC checklist satisfied and the statement "The authors declare that they have no competing interests" added.

Tables - Shading deleted and replaced with bold to highlight the appropriate cells.

Tables - All borders of tables now fully visible as black lines.

Terry N Flynn
Tim J Peters